Author's response to reviews

Title: Fatal poisonings in Oslo: A one-year observational study

Authors:

Mari A Bjornaas (mabjornaas@gmail.com)
Brita B Teige (brita.teige@medisin.uio.no)
Knut KE Hovda (kehovda@yahoo.no)
Oivind O Ekeberg (oivind.ekeberg@uus.no)
Fridtjof F Heyerdahl (fridtjof.heygerdahl@medisin.uio.no)
Dag D Jacobsen (daja@uus.no)

Version: 3 Date: 24 May 2010

Author's response to reviews:

This is the proofreading stage. I have completed the major revisions required, and most of the minor revisions. Unfortunately, I was not able to crop the figure as requested, and ask for your assistance to do this.

Yours sincerely,
Dr Bjornaas